Dear Editors,

Thank you very much for the careful review and valuable suggestions concerning our manuscript (70233). We have revised the manuscript according to the reviewers’ comments. The point-by-point response to the reviewers’ evaluation is below.

1. Background in ABSTRACT can mention that this case was followed up to 5 years.

Response to comments #1

Thanks very much for your constructive comments. We have changed it to a new citation.

2. Chief complaints need to specify in-patient unit or ICU or out-patient clinic.

Response to comments #2

We are grateful for your valuable suggestions. The patient was admitted to our outpatient clinic.

3. The figures were not clear, please provide clearer figures.

Response to comments #3

We are grateful for your valuable suggestions. We replace them with
clearer figures.

4. Language Quality

Thanks very much for your constructive comments. We made some serious changes to the language.